Author’s response to reviews

Title: Pancreatic adenocarcinoma exerts systemic effects on the peripheral blood myeloid and plasmacytoid dendritic cells: an indicator of disease severity?

Authors:

Vegard Tjomsland (Vegard.Tjomsland@liu.se)
Per Sandström (Per.Sandstrom@liu.se)
Anna Spångeus (anna.spangeus@liu.se)
Davorka Messmer (dmessmer@ucsd.edu)
Johan Emilsson (johem234@student.liu.se)
Ursula Falkmer (Ursula.Falkmer@lj.se)
Sture Falkmer (Sture.Falkmer@lj.se)
Karl-Eric Magnusson (karl-eric.magnusson@liu.se)
Kurt Borch (Kurt.Borch@liu.se)
Marie Larsson (marie.larsson@liu.se)

Version: 4 Date: 6 March 2010

Author’s response to reviews:

BMC Cancer
Dear Editor,

My colleagues and I would like to resubmit our revised manuscript 3702671363140567 entitled “Pancreatic adenocarcinoma exerts systemic effects on the peripheral blood myeloid and plasmacytoid dendritic cells: an indicator of disease severity?” for consideration for publication in BMC Cancer.

Formatting the manuscript

Major revisions:
– Figure 4 is now one file.
The figures are cited in order Figure 1, Figure 2 etc. and all figures are cited in the manuscript.
– The figures are now listed without ranges e.g. Figure A, B, C, and D instead of Figure A-D. Please let me know if this was the ranges referred to in the comment.
– All figures were listed
– All mention of figures is now consistent using Fig in the result section

Minor revisions
– All highlighting/tracking is removed from the manuscript
– The figures are cropped as closely as possible
– The manuscript is checked for typographical errors

Thank you for your consideration.

Marie Larsson
Marie Larsson, Ph.D. Associate Professor
Molecular Virology
Linköping University
581 85 Linköping
Sweden
Email: marie.larsson@liu.se
Phone +46 13 221055 +46 07 3658050